

# **CLINICAL MEMORANDUM**

From: Carlos H. Villa MD PhD, Medical Officer, OBRR/DBCD/CRS

To: Nicole Verdun MD, Director, OBRR

Through: Orieji Illoh MD, Division Director, OBRR/DBCD

Wendy Paul MD, Deputy Division Director, OBRR/DBCD

Salim Haddad MD, Team Lead, OBRR/DBCD/CRS

Re: EUA 26382

Product: COVID-19 Convalescent Plasma

Sponsor: Assistant Secretary for Preparedness and Response (ASPR)

Office of Assistant Secretary for Preparedness and Response (ASPR)

U.S. Department of Health and Human Services (HHS)

#### **EXECUTIVE SUMMARY**

COVID-19 Convalescent Plasma (CCP) was granted Emergency Use Authorization (EUA) for the treatment of hospitalized patients with COVID-19 on August 23, 2020. At that time, the totality of the scientific evidence supported a determination that the use of CCP in hospitalized patients with COVID-19 met the "may be effective" standard for issuance of an EUA, and it was reasonable to consider that the known and potential benefits of CCP outweighed the known and potential risks for the authorized use.

Following the EUA, emerging evidence from randomized controlled trials has continued to inform this assessment. The available evidence indicates that 1) the use of low titer CCP in hospitalized patients no longer meets the evidentiary standard of "may be effective", and 2) high titer CCP has not demonstrated benefit when administered late in the disease course in immunocompetent hospitalized patients.

Additional data from RCTs and observational studies support a determination that high titer CCP may be effective when administered early in the course of illness, particularly prior to the expected time of host antibody response. In patients with impaired humoral immunity, the potential therapeutic window may be prolonged. As RCT data continue to demonstrate that the risks of CCP do not exceed those observed with plasma transfusion in general, it is reasonable to believe that the known and potential benefits of use of high titer CCP outweigh the known and potential risks when used for the treatment of hospitalized patients with COVID-19, particularly early in the course of illness or in patients with impaired humoral immunity. Results from ongoing RCTs are expected to continue inform the optimal patient populations and product characteristics for efficacy of CCP in COVID-19.

**Recommendation:** The conditions of Emergency Use Authorization for CCP for the treatment of hospitalized patients with COVID-19 should be revised to exclude the use of low titer CCP. High titer CCP continues to meet criteria for Emergency Use Authorization for the treatment of hospitalized patients with COVID-19 early in the course of hospitalization. The letter of authorization and accompanying fact sheets should be revised to emphasize early administration of CCP in the absence of immunodeficiency or immunosuppression. Additional results from adequate and well-controlled trials will continue to be evaluated to further characterize the patient populations and product characteristics meeting EUA criteria.

### **Regulatory History**

FDA issued an Emergency Use Authorization (EUA 26382) on August 23, 2020 for the use of COVID-19 Convalescent Plasma (CCP) for treatment of hospitalized patients with COVID-19. This authorization was based on the totality of the scientific evidence available at the time, which supported a determination that CCP met the "may be effective" criterion for issuance of an EUA and that the known and potential benefits of CCP outweighed the known and potential risks of CCP for the terms of the EUA. Considering the limited data from adequate and well-controlled randomized trials at the time of the issuance of the EUA, FDA noted in the August 23, 2020 Letter of Authorization that additional data from such trials remained necessary for a definitive demonstration of CCP efficacy and to determine the optimal product attributes and appropriate patient populations for use of CCP. Information derived from ongoing clinical trials of CCP, particularly randomized, controlled trials, as well as clinical trial results from studies of other investigational medical products to treat COVID-19, were expected to inform the continuing risk benefit assessment. The current proposed amendment to the EUA revises the conditions of authorization to reflect the accumulated evidence on the use of CCP in COVID-19.

# Summary of Evidence Following the August 23, 2020 EUA

Randomized Controlled Trials (RCTs)

At the time of the issuance of the original EUA for CCP, results from two RCTs had been published or made publicly available[1, 2]. Both studies failed to demonstrate a significant benefit with CCP transfusion, but may have been underpowered due to early stopping either due to low enrollment[1] or to the presence of high antibody titers in subjects prior to transfusion[2]. Additionally, patients were treated relatively late in illness, at a median of 30 days[1] and 10 days[2] post symptom onset. Following issuance of the EUA for CCP, the results of several additional RCTs of CCP were published or made publicly available.

The PLACID study, conducted in India, did not observe a difference in progression to severe disease or death by 28 days in 235 patients receiving CCP and standard of care compared to 229 subjects randomized to standard of care alone (open label)[3]. At enrollment, subjects reported a median of 8 days of symptoms (IQR 6-11) and 83% had detectable neutralizing antibodies. A higher proportion of patients in the interventional arm showed resolution of shortness of breath and fatigue on day 7, and negative conversion of SARS-CoV-2 RNA at day 7 was higher in the intervention arm. Other secondary outcomes (including days of respiratory support) showed no significant differences. The findings of the trial may have been limited by the relatively low titers in the transfused plasma. Only 71.4% of participants received plasma with detectable neutralizing antibodies and donors had a median titer of 1:40 (IQR 1:30-1:80) compared to a median titer of 1:90 (IQR 1:30-1:240) in recipients at enrollment. While subgroup analyses based on titers and duration of symptoms did not demonstrate significant differences, these may have been underpowered to detect clinically important differences. For example, the subgroup analysis for those with symptoms for three days or less at enrollment was based on only 24 subjects and found a risk ratio for the composite outcome of 0.8 (95% confidence interval 0.2 to 3.1).

The PlasmAr study in Argentina, a double-blind placebo-controlled study of CCP in hospitalized patients with severe COVID-19, found no improvement at 30 days in clinical status or overall mortality with transfusion of 500 mL of CCP in 228 test subjects compared to 105 control subjects receiving a normal saline placebo[4]. Prior to transfusion of CCP, titers were measured using an ELISA assay for antibodies against the SARS-CoV-2 spike protein and receptor binding domain and showed a median titer of 1:3200 (IQR 1:800-1:3200). Based on the correlation of this assay with neutralization activity, these titers would be consistent with a classification of high titer under this EUA (ID50 neutralization titer >1:250). The median time from symptom onset to enrollment was 8 days (IQR 5-11) and 54% of subjects had detectable antibodies at baseline. The prespecified subgroup analyses failed to demonstrate credible subgroup effects but was limited by the relatively small subgroup size (for example, 39 subjects [17%] received the intervention within 72 hours of symptom onset). No significant differences in the overall incidence of adverse events were observed, and the most common infusion-related event was nonhemolytic febrile reactions.

The ConPlas-19 study in Spain observed a trend towards a survival benefit with administration of high titer CCP (median neutralization titer 1:292, IQR 238-451) to hospitalized patients with COVID-19, but the difference did not meet the prespecified statistical test for significance[5], and the study was stopped early due to diminished enrollment as the local number of cases dropped, resulting in a small number of events and limited statistical power. 81 patients were randomized to the study and reported a median of 8 days of symptoms (IQR 6-9) with 49.4% positive for anti-SARS-CoV-2 antibodies at enrollment. No significant differences were found in secondary endpoints, including time to improvement on an ordinal scale of severity, proportion of patients requiring high flow oxygen, non-invasive ventilation, or higher respiratory support, hospital length of stay, and days free from oxygen support.

A small randomized controlled trial of CCP in patients with severe COVID-19 compared to standard of care (n=40 per arm) found no significant clinical benefit in the overall population, although a post-hoc subgroup of patients younger than 67 years old demonstrated clinical benefit[6]. Plasma was tested for the presence of neutralizing antibodies using a surrogate neutralization assay, but further quantitative information on antibody content or titer was not provided. Another small randomized pilot study (n=20 per arm) found CCP transfusion to be safe but also did not observe differences in the primary outcome of mechanical ventilation compared to standard of care[7]. Prospective antibody tittering was not performed.

The RECOVERY study in the United Kingdom, a large platform trial that has evaluated multiple COVID-19 therapies[8], included a comparison of CCP to standard of care. A January 15, 2021 statement from the RECOVERY trial chief investigators reported that enrollment in the CCP arm would be halted after the independent Data Monitoring Committee saw no convincing evidence that further recruitment would provide conclusive proof of worthwhile mortality benefit[9]. The preliminary analysis, based on 1873 reported deaths among 10,406 randomized patients, showed no significant difference in the primary endpoint of 28-day mortality. The results of the study have not been published at this time, and additional information on the timing of product administration relative to symptom onset, baseline patient antibody titers, and antibody titers in the transfused CCP will be important to further evaluate the findings.

In the outpatient setting, Libster et al reported the results of a randomized, double-blind, placebocontrolled trial of 250 mL of CCP versus saline placebo in 160 high risk adults (75 years or older, or 65 to 74 with at least one comorbidity (hypertension, diabetes, obesity, chronic renal failure, cardiovascular disease, and COPD)[10] with mild COVID-19 symptoms. The investigators used the same serologic assay used to qualify CCP in the PlasmAr study described above and used plasma from donors with a titer of 1:1000 or greater. The intervention was administered less than 72 hours after the onset of symptoms and the primary endpoint was progression to severe respiratory disease. While the trial was halted due to a local decrease in cases, at 76% of target enrollment, they observed a 48% relative risk reduction in development of severe respiratory disease within 15 days, with the primary outcome occurring in 25 of 80 subjects (31%) in the placebo arm, and 13 of 80 subjects (16%) in the CCP arm (relative risk, 0.52; 95% CI, 0.29 to 0.94; p=0.03). A modified intent-to-treat analysis excluding patients who experienced the primary end point prior to the study intervention found a relative risk of 0.4 (95% CI 0.20 to 0.81). While secondary end points (including life-threatening respiratory disease, oxygen supplementation at 100% FiO2, noninvasive ventilation, ICU admission, mechanical ventilation, critical systemic illness, and acute respiratory failure) all trended towards benefit, differences were not statistically significant, with wide confidence intervals due to the small number of these events across both groups. A dose dependent effect was described, with a relative risk of 0.27 (95% CI 0.08-0.68) above a median titer of 1:3200 and a relative risk reduction of 0.69 (95% CI 0.34-1.31) below the median of 1:3200.

#### Observational Studies

Some observational and non-randomized studies published following the original EUA have shown potential benefit with early, high-titer CCP, or in specific subgroups, but findings have been variable, and minimal or no benefit has been seen in late disease once respiratory failure has progressed to the stage of requiring mechanical ventilation/intubation.

In an updated observational study using propensity score matching to non-transfused control patients, Salazar et al. observed improved mortality in subjects transfused within 72 hours of admission with high-titer CCP, but not in subjects transfused beyond 72 hours[11]. Additional analysis found an optimal window for transfusion within 44 hours of hospital admission for discriminating mortality within 60 days after transfusion.

An observational study comparing 263 patients hospitalized with severe COVID-19 treated under the EAP to 263 matched controls[12], found no statistical difference in 28-day mortality CCP cases (25.5%) compared to controls (27%,P = 0.06). 7-day mortality was statistically better for CCP cases (9.1%) than controls (19.8%, P < 0.001) and continued at 14 days (14.8% vs.23.6%, P = 0.01). Titers of the CCP transfused (1-2 units) were not available.

A non-randomized cohort study in Kuwait, in patients with moderate or severe COVID-19, found that CCP transfusion in 135 subjects was associated with a higher rate of clinical improvement compared to 233 matched controls receiving standard of care[13], but quantitative titers of the transfused CCP were not available.

A matched cohort analysis found no significant difference in survival in 64 patients who received CCP a median of 7 days after symptom onset compared to 177 matched controls[14].

An observational study of CCP transfused to 73 hospitalized COVID-19 patients within 72 hours of admission found no overall difference in mortality or oxygenation compared to propensity-score matched controls, although there was evidence of a mortality and oxygenation benefit in CCP recipients <65 years old[15]. In CCP recipients, pre-transfusion antibody titers were associated with mortality at day 28 in univariate analyses.

A recent observational case-control study found that hospitalized COVID-19 patients with severe respiratory failure who were transfused with CCP presented with already high titers of SARS-CoV-2 IgG antibodies before transfusion, and did not show improved survival at 30 days compared to matched controls[16].

Expanding on analyses made publicly available prior to the original EUA[17], a recent report from the national Expanded Access Protocol (EAP) sponsored by the Mayo Clinic described a survival benefit at 30 days following transfusion associated with early use of high titer CCP in hospitalized patients[18], where titer was measured by the Ortho VITROS IgG assay included as a CCP manufacturing test under this EUA. FDA has performed additional analyses of over 20,000 subjects enrolled in the EAP using multivariate Cox proportional-hazards regression that included titer (<250 versus >250 in a live virus neutralization assay), ventilation status, age, days from diagnosis to transfusion, gender, race, and hospital region as covariates. In findings similar to those at time of EUA issuance, there was a modest survival benefit associated with transfusion of high titer plasma (HR 0.93 [0.88,0.98] for high versus low titer) and with earlier transfusion (transfusion on same day of diagnosis: HR 1 [reference]; 1 to 3 days from diagnosis: HR 1.08 [0.93, 1.26]; 4 to 10 days from diagnosis: HR 1.23 [1.06, 1.43]; 11 days or more diagnosis: HR 1.23 [1.05, 1.44]). Patients aged 41-60 years (HR 1.53 [1.32, 1.78]), 61-80 years (HR 2.90 [2.50, 3.35]), and 81 years or older (HR 5.74 [4.92, 6.69]) were all more likely to experience death when compared to patients 40 years or younger, with an increasing hazard with each successive age category. Male sex was also associated with an increased hazard (HR 1.11 [1.05, 1.17]).

Finally, several reports of clinical improvement in immunosuppressed or immunodeficient patients treated with CCP, including after prolonged illness, support potential efficacy in this population and suggest a longer potential therapeutic window than in immunocompetent patients, but well-controlled data in these populations remain lacking[19-21].

#### Preclinical studies

Early preclinical studies using mouse and hamster models suggested potential benefit with passive immune therapies. More recently, macaque models found that passive transfer of immunoglobulins could provide protection against infection, as well as therapeutic efficacy following viral challenge, in a dose-dependent manner[22]. Higher doses were needed for therapeutic versus prophylactic efficacy. While the authors noted that their data demonstrated the therapeutic efficacy of convalescent plasma for treatment of infection, they cautioned that only high serum neutralizing antibody titers showed therapeutic benefit, potentially at levels that would be difficult to achieve via plasma transfusion.

The relative roles of humoral and cellular immunity in SARS-CoV-2 infection continue to be unraveled, and it appears likely that CD4+ T cells, CD8+ T cells, and neutralizing antibody responses all contribute to control of SARS-CoV-2 infection in both non-hospitalized and hospitalized cases of COVID-19[23]. The large majority of patients with SARS-CoV-2 infection will seroconvert within 5-15 days post-symptom onset, with 90% seroconverting by day 10[23-25]. IgM and IgG antibodies are frequently detected concurrently[26], and peak anti-spike or anti-RBD IgG levels are reached by approximately 15 days post symptom onset[27]. Antibody responses and memory B cells appear to persist for at least 5 months and antibodies may be a correlate of immune protection[28-31]. Delayed antibody response kinetics also appear to be associated with more severe disease[27, 32]. At the same time, studies have generally shown higher titers in patients following recovery from severe disease compared to mild or asymptomatic illness[25, 33].

The observation that high titer CCP was beneficial when administered within 72 hours of symptom onset in high risk subjects, but failed to demonstrate benefit in trials where the median duration of symptoms was 8 days or longer, indicates benefit with CCP transfusion is more likely in patients early in the humoral immune response when host antibody titers remain undetectable or low (i.e., likely within the first week following symptom onset). This is consistent with longstanding historical precedent in passive immune therapies for viral infections, where prophylactic or early use has generally been more effective than in established infections[34].

These trends are also consistent with clinical evidence for administration of anti-SARS-CoV-2 monoclonal antibodies, where benefit has been demonstrated with early outpatient use, but not in hospitalized patients within 12 days of symptom onset[35-37] as described in the following two studies:

In outpatient studies of bamlanivimab in recently diagnosed patients with mild to moderate disease (BLAZE-1)[36], subjects were excluded if they were previously known to be seropositive. Subjects had a median of 4 days of symptoms at the time of infusion, and the study found one of three doses of neutralizing antibody LY-CoV555 appeared to accelerate the natural decline in viral load over time. While reduction in viral load was the primary endpoint in this phase 2 trial, subjects treated with bamlanivimab also showed a nominally statistically significant reduction in COVID-19 related hospitalizations or ED visits within 28 days in the pooled dose-level data.

In outpatient studies of casirivimab/imdevimab in symptomatic patients with mild to moderate COVID-19 (R10933-10987-COV-2067), subjects who were no more than 7 days from symptom enrollment were included regardless of serostatus[35]. Casirivimab/imdevimab treatment reduced viral load, and patients who were seronegative at baseline showed larger reductions in viral load and a larger reduction in the proportion of subjects with at least one medically attended visit compared to the overall population. Based on these studies, both therapies were granted EUA for use in high risk outpatients with mild to moderate COVID-19 (https://www.fda.gov/media/143892/download, https://www.fda.gov/media/143603/download).

In early studies of the COVID-19 pandemic, the median time from symptom onset to the development of dyspnea was approximately 5-8 days[38, 39], and patients who develop critical illness typically do so shortly thereafter (days 8-10)[40]. While the study by Libster et al[10] demonstrated a reduction in progression to severe disease in high risk outpatients within 72 hours of the onset of symptoms, one factor complicating very early use of CCP in the outpatient population is the evidence that a large proportion of these patients will have a self-limited illness and will not go on to severe or critical illness even without targeted intervention[41]. Therefore, in the early-disease outpatient population, it is important to have a full understanding of the relative benefit and identify high-risk populations so that the known and potential risks of transfusion are outweighed by the known and potential benefits of CCP. Ongoing randomized controlled trials will be critical to determine the clinical and laboratory parameters that can identify where the potential benefit of CCP outweighs the potential risk in outpatients.

Based on the study by Libster et al[10] the therapeutic window appears to be at least within 72 hours of symptom onset, while additional negative RCTs with a median duration of symptoms prior to transfusion of 8 days indicating that 8 days after symptom onset may be too late for efficacy of CCP in immunocompetent hospitalized COVID-19 patients. These timepoints appear to correlate with the timing of the patients' own antibody responses to infection, such that by the time a patient is forming their own antibodies, benefit from CCP appears unlikely. The time period between 3 and 7 days remains to be studied rigorously in randomized trials of CCP, but observational studies, preclinical studies, studies of related therapies, and what is known about the timing of the adaptive immune response in SARS-CoV-2 infection suggest that high titer CCP may be effective in this window period. As noted above, this window appears to be longer in the setting of impaired or deficient humoral immunity. Nonetheless, adequate and well-controlled trials in this time period remain necessary for a conclusive demonstration of efficacy.

#### **Evaluation of EUA Criteria**

Considering the emerging evidence summarized above, the EUA criteria based on FDA Guidance<sup>1</sup> were reevaluated to determine the ongoing eligibility of CCP for the treatment of hospitalized patients with COVID-19.

# a. Serious or life-threatening disease or condition

COVID-19 is a serious or life-threatening disease or condition which has resulted in >380,000 deaths in the United States as of January 15, 2021, and large numbers of new infections and deaths continue to be reported (www.cdc.gov/coronavirus/2019-ncov/cases-updates/us-cases-deaths.html). Patients also have an increased risk of serious events such as thromboembolic events, cardiomyopathy and arrhythmia, renal injury, and stroke, which can result in long-term morbidity (https://www.cdc.gov/coronavirus/2019-ncov/hcp/clinical-guidance-management-patients.html). Therefore, COVID-19 continues to meet the criterion of a serious or life-threatening disease or condition.

# b. Evidence of Effectiveness

Based on newly available evidence from randomized, controlled trials, observational studies, and preclinical studies summarized above, use of low titer CCP no longer meets the evidentiary standard of "may be effective". Available evidence supports that high titer CCP is effective when administered within 3 days of symptom onset but is unlikely to be effective when administered 8 days or more following symptom onset in immunocompetent patients. The time period between 3 days and 7 days post symptom onset remains to be rigorously studied in randomized, controlled trials. However, in aggregate, the available data suggest a window for potential therapeutic efficacy if CCP is administered early (typically less than 7 days of symptom onset), with benefit most likely the earlier the CCP is transfused. Additional clinical and preclinical evidence on the role and timing of the humoral immune response COVID-19 summarized above are consistent with this being the period prior to a patient's own antibody response.

In early reports from Wuhan, China, the median time from symptom onset to hospitalization was reported to be 7 days[38]. However, additional reports showed this time period shortened as the pandemic progressed and was as short as 1.5 days as awareness of SARS-CoV-2 increased[42]. Similarly, in an early US case series, the mean duration of symptoms at admission was 7±4 days[43], but a later report of 463 patients with COVID-19 in Detroit, MI, found 79-85% of patients had symptom duration of <7 days prior to admission[44]. Significant variability between countries and patient demographics has also been observed[45]. Considering the reported timing of hospitalization and time course of COVID-19, while highly variable, there appears to be a potential therapeutic window if CCP is administered to hospitalized patients early in the course of hospitalization. Recognizing the variability in patients' courses of illness[46], underlying immune function, kinetics of antibody response, stage of disease at presentation to care, and CCP potency, a precise therapeutic window is difficult to define. Therefore, it is reasonable to consider that treatment of hospitalized COVID-19 patients continues to meet the standard of "may be effective", although use early in the course of hospitalization should be emphasized over treatment of late, severe disease (e.g., in intubated and mechanically ventilated patients). In addition, use of high titer CCP in patients with impaired humoral immunity, including later in illness, appears to be associated with clinical improvement and meets the evidentiary standard of "may be effective" although the quality of evidence in these populations remains limited.

Several studies have failed to demonstrate benefit in intubated patients requiring mechanical ventilation, but these patients have generally been late in the time course of illness (>8-10 days). While expected to be rare, patients whose progression of disease requires intubation early in the course of illness, or who may have been intubated early for other pre-existing clinical reasons (e.g., acquired infection while already intubated, intubated for risk factors related to airway protection rather than ARDS), may still be in the early phase of the humoral immune response (i.e., prior to formation of neutralizing antibodies), and could potentially benefit from passive immune therapy, including CCP transfusion. Also, as noted above, patients with impaired humoral immunity may have a longer therapeutic window for efficacy, including following intubation and mechanical ventilation. Therefore, an exclusion based solely on intubation status does not appear justified at this time, especially considering the acceptable safety profile of CCP demonstrated to date.

Ongoing trials in diverse clinical settings, including multicenter outpatient and inpatient clinical studies, will be critical to the continued evaluation of safety and efficacy of CCP for the treatment of COVID-19. Patients should continue to be encouraged to enroll in randomized, controlled trials when available.

# c. Risk-Benefit Analysis

Potential benefits of CCP as outlined above and in the prior EUA include reduced progression to severe disease, improved mortality with early treatment, and improved viral clearance.

Risks are expected to include those inherent to plasma transfusion[47]:

- o Transfusion related acute lung injury (TRALI)
- Transfusion associated cardiac overload (TACO)
- o Allergic/Anaphylactic reactions
- o Febrile nonhemolytic transfusion reactions
- Transfusion-transmitted infections
- Hemolytic reactions

The risks of these events observed in trials to date appear to be within the expected rates of these events for transfusion of plasma in critically ill patients[48-50].

Additional risks specific to convalescent plasma include a theoretical risk of antibody-dependent enhancement (ADE) and a theoretical risk of suppressed long-term immunity. Antibody-dependent enhancement of disease is thought to occur when antibodies to an infectious agent 'bridge' the pathogen to Fc receptors on immune cells leading to increased viral entry and enhancement of infection[51]. No evidence of antibody-dependent enhancement of disease has been observed in the studies of CCP summarized above, and the potential for ADE with the use of CCP remains theoretical at this time.

The potential of passive immune therapies to suppress long-term immunity in recovered or vaccinated patients has not been evaluated in clinical studies to date. Current CDC guidance recommends that patients who receive CCP for the treatment of COVID-19 defer vaccination for at least 90 days as a precautionary measure until additional information becomes available, to avoid potential interference of the antibody therapy with vaccine-induced immune responses (https://www.cdc.gov/vaccines/covid-19/info-by-product/clinical-considerations.html). Ongoing trials will evaluate antibody responses to infection following treatment with CCP.

# **Summary of Risk-Benefit Analysis**

The available evidence indicates a lack of benefit with the use of low titer CCP in the treatment of COVID-19. Therefore, considering the known risks of plasma transfusion outlined above, the

known and potential benefits of low titer CCP use no longer outweigh the known and potential risks.

Additional RCTs to date have demonstrated benefit with transfusion of high titer CCP within 3 days of symptom onset, and a lack of benefit in hospitalized COVID-19 patients in studies at a median of 8 days or later post symptom onset. These findings suggest CCP is less likely to be effective when the host humoral response has already been established. While the study by Libster et al demonstrated benefit in the outpatient setting, the known and potential benefits in the outpatient setting when disease is still mild must be weighed against known and potential risks of plasma transfusion. Because many patients very early in disease (within 72 hours of symptoms) will have mild to moderate self-limited disease, and adverse events associated with plasma transfusion can be serious or even fatal, additional data from randomized controlled trials in the outpatient setting are needed to assure that known and potential risks are outweighed by known and potential benefits in outpatients. Such studies are underway and will continue to inform this risk-benefit analysis.

As noted above, based on results from RCTs and the EAP, the safety profile of CCP appears comparable to that of plasma transfusion in general. Considering this safety profile, it is reasonable to believe that the known and potential benefits of high titer CCP outweigh the known and potential risks when used for the early treatment of hospitalized COVID-19 patients. Based on what is known about the typical course of illness and kinetics of the humoral immune response in COVID-19, for most patients, early treatment likely represents within one week of symptom onset. However, considering the large variability in the course of illness in COVID-19, and the variability in host immune function and kinetics, a specific timeframe for efficacy cannot be defined based on available data to date. Furthermore, the therapeutic window may be longer when CCP is administered to patients with clinical or laboratory evidence of impaired humoral immunity.

#### d. No alternatives

While several therapeutics have been granted emergency use authorization, only Remdesivir has been approved for the treatment of COVID-19. However, for the purposes of determining the eligibility of CCP for EUA under section 564 of the Federal Food, Drug, and Cosmetic Act (21 USC 360bbb-3), Remdesivir is not considered an adequate alternative for the treatment of hospitalized patients with COVID-19. Therefore, CCP continues to meet this criterion because there are currently no adequate, approved, and available alternatives to CCP for the treatment of hospitalized patients with COVID-19.

In sum, the EUA for CCP should be revised so that the eligibility criteria for Emergency Use Authorization outlined in FDA guidance¹ continue to be met. The conditions of Emergency Use Authorization for CCP in hospitalized patients with COVID-19 should be revised to exclude the use of low titer CCP. High titer CCP continues to meet criteria for Emergency Use Authorization for the treatment of hospitalized patients with COVID-19 early in the course of hospitalization, and the letter of authorization and accompanying fact sheets should be revised to emphasize early transfusion in the absence of immunodeficiency or immunosuppression. Additional results from adequate and well-controlled trials will continue to be evaluated to further characterize the patient populations and product

### References

- 1. Li, L., et al., Effect of Convalescent Plasma Therapy on Time to Clinical Improvement in Patients With Severe and Life-threatening COVID-19: A Randomized Clinical Trial. JAMA, 2020.
- 2. Gharbharan, A., et al., *Convalescent Plasma for COVID-19. A randomized clinical trial.* medRxiv, 2020: p. 2020.07.01.20139857.
- 3. Agarwal, A., et al., Convalescent plasma in the management of moderate covid-19 in adults in India: open label phase II multicentre randomised controlled trial (PLACID Trial). BMJ, 2020. **371**: p. m3939.
- 4. Simonovich, V.A., et al., *A Randomized Trial of Convalescent Plasma in Covid-19 Severe Pneumonia*. New England Journal of Medicine, 2020.
- 5. Avendano-Sola, C., et al., *Convalescent Plasma for COVID-19: A multicenter, randomized clinical trial.* medRxiv, 2020: p. 2020.08.26.20182444.
- 6. Ray, Y., et al., Clinical and immunological benefits of convalescent plasma therapy in severe COVID-19: insights from a single center open label randomised control trial. medRxiv, 2020: p. 2020.11.25.20237883.
- 7. AlQahtani, M., et al., Randomized controlled trial of convalescent plasma therapy against standard therapy in patients with severe COVID-19 disease. medRxiv, 2020: p. 2020.11.02.20224303.
- 8. Normand, S.-L.T., *The RECOVERY Platform*. New England Journal of Medicine, 2020.
- 9. Statement from RECOVERY trial chief investigators. 2021. Available from: <a href="https://www.recoverytrial.net/news/statement-from-the-recovery-trial-chief-investigators-15-january-2021-recovery-trial-closes-recruitment-to-convalescent-plasma-treatment-for-patients-hospitalised-with-covid-19.">https://www.recoverytrial.net/news/statement-from-the-recovery-trial-chief-investigators-15-january-2021-recovery-trial-closes-recruitment-to-convalescent-plasma-treatment-for-patients-hospitalised-with-covid-19.</a>
- 10. Libster, R., et al., Early High-Titer Plasma Therapy to Prevent Severe Covid-19 in Older Adults. New England Journal of Medicine, 2021.
- 11. Salazar, E., et al., Significantly Decreased Mortality in a Large Cohort of Coronavirus Disease 2019 (COVID-19) Patients Transfused Early with Convalescent Plasma Containing High-Titer Anti-Severe Acute Respiratory Syndrome Coronavirus 2 (SARS-CoV-2) Spike Protein IgG. The American journal of pathology, 2021. 191(1): p. 90-107.
- 12. Shenoy, A.G., et al., *Early mortality benefit with COVID-19 convalescent plasma: a matched control study.* British Journal of Haematology. **n/a**(n/a).
- 13. Alsharidah, S., et al., COVID-19 Convalescent Plasma Treatment of Moderate and Severe Cases of SARS-CoV-2 Infection: A Multicenter Interventional Study. International Journal of Infectious Diseases, 2020.
- 14. Rogers, R., et al., *Convalescent plasma for patients with severe COVID-19: a matched cohort study.* Clinical Infectious Diseases, 2020.
- 15. Yoon, H.a., et al., *Treatment of Severe COVID-19 with Convalescent Plasma in the Bronx, NYC.* medRxiv, 2020: p. 2020.12.02.20242909.
- 16. Klein, M.N., et al., *Kinetics of SARS-CoV-2 antibody responses pre- and post- COVID-19 convalescent plasma transfusion in patients with severe respiratory failure: an observational case-control study.* medRxiv, 2020: p. 2020.12.10.20247007.
- 17. Joyner, M.J., et al., Effect of Convalescent Plasma on Mortality among Hospitalized Patients with COVID-19: Initial Three-Month Experience. medRxiv, 2020.

- 18. Joyner, M.J., et al., *Convalescent Plasma Antibody Levels and the Risk of Death from Covid-19*. New England Journal of Medicine, 2021.
- 19. Senefeld, J.W., et al., *Therapeutic use of convalescent plasma in COVID-19 patients with immunodeficiency*. medRxiv, 2020: p. 2020.11.08.20224790.
- 20. Clark, E., et al., Convalescent plasma for persisting COVID-19 following therapeutic lymphocyte depletion: a report of rapid recovery. British Journal of Haematology, 2020. **190**(3): p. e154-e156.
- 21. Ferrari, S., et al., Convalescent hyperimmune plasma for chemo-immunotherapy induced immunodeficiency in COVID-19 patients with hematological malignancies. Leuk Lymphoma, 2021: p. 1-9.
- 22. McMahan, K., et al., *Correlates of protection against SARS-CoV-2 in rhesus macaques*. Nature, 2020.
- 23. Sette, A. and S. Crotty, Adaptive immunity to SARS-CoV-2 and COVID-19. Cell.
- 24. Wajnberg, A., et al., *Humoral response and PCR positivity in patients with COVID-19 in the New York City region, USA: an observational study.* Lancet Microbe, 2020. **1**(7): p. e283-e289.
- 25. Röltgen, K., et al., Defining the features and duration of antibody responses to SARS-CoV-2 infection associated with disease severity and outcome. Science Immunology, 2020. 5(54): p. eabe0240.
- 26. Long, Q.-X., et al., *Antibody responses to SARS-CoV-2 in patients with COVID-19*. Nature Medicine, 2020. **26**(6): p. 845-848.
- 27. Lucas, C., et al., *Kinetics of antibody responses dictate COVID-19 outcome*. medRxiv, 2020: p. 2020.12.18.20248331.
- 28. Gaebler, C., et al., Evolution of antibody immunity to SARS-CoV-2. Nature, 2021.
- 29. Wajnberg, A., et al., *Robust neutralizing antibodies to SARS-CoV-2 infection persist for months.* Science, 2020. **370**(6521): p. 1227-1230.
- 30. Lumley, S.F., et al., *Antibody Status and Incidence of SARS-CoV-2 Infection in Health Care Workers.* New England Journal of Medicine, 2020.
- 31. Letizia, A.G., et al., SARS-CoV-2 seropositivity and subsequent infection risk in healthy young adults: a prospective cohort study. medRxiv, 2021: p. 2021.01.26.21250535.
- 32. Ren, L., et al., *The kinetics of humoral response and its relationship with the disease severity in COVID-19.* Communications Biology, 2020. **3**(1): p. 780.
- 33. Robbiani, D.F., et al., *Convergent antibody responses to SARS-CoV-2 in convalescent individuals*. Nature, 2020. **584**(7821): p. 437-442.
- 34. Casadevall, A., E. Dadachova, and L.-a. Pirofski, *Passive antibody therapy for infectious diseases*. Nature Reviews Microbiology, 2004. **2**(9): p. 695-703.
- 35. Weinreich, D.M., et al., *REGN-COV2*, a Neutralizing Antibody Cocktail, in Outpatients with Covid-19. New England Journal of Medicine, 2020.
- 36. Chen, P., et al., *SARS-CoV-2 Neutralizing Antibody LY-CoV555 in Outpatients with Covid-19*. New England Journal of Medicine, 2020.
- 37. A Neutralizing Monoclonal Antibody for Hospitalized Patients with Covid-19. New England Journal of Medicine, 2020.
- 38. Wang, D., et al., Clinical Characteristics of 138 Hospitalized Patients With 2019 Novel Coronavirus–Infected Pneumonia in Wuhan, China. JAMA, 2020. **323**(11): p. 1061-1069.

- 39. Huang, C., et al., Clinical features of patients infected with 2019 novel coronavirus in Wuhan, China. Lancet, 2020. **395**(10223): p. 497-506.
- 40. Berlin, D.A., R.M. Gulick, and F.J. Martinez, *Severe Covid-19*. New England Journal of Medicine, 2020. **383**(25): p. 2451-2460.
- 41. Wu, Z. and J.M. McGoogan, Characteristics of and Important Lessons From the Coronavirus Disease 2019 (COVID-19) Outbreak in China: Summary of a Report of 72 314 Cases From the Chinese Center for Disease Control and Prevention. JAMA, 2020. **323**(13): p. 1239-1242.
- 42. Sanche, S., et al., *High Contagiousness and Rapid Spread of Severe Acute Respiratory Syndrome Coronavirus* 2. Emerg Infect Dis, 2020. **26**(7): p. 1470-1477.
- 43. Bhatraju, P.K., et al., *Covid-19 in Critically Ill Patients in the Seattle Region Case Series.* New England Journal of Medicine, 2020. **382**(21): p. 2012-2022.
- 44. Suleyman, G., et al., Clinical Characteristics and Morbidity Associated With Coronavirus Disease 2019 in a Series of Patients in Metropolitan Detroit. JAMA Netw Open, 2020. **3**(6): p. e2012270.
- 45. Faes, C., et al., *Time between Symptom Onset, Hospitalisation and Recovery or Death:* Statistical Analysis of Belgian COVID-19 Patients. Int J Environ Res Public Health, 2020. **17**(20).
- 46. Argenziano, M.G., et al., *Characterization and clinical course of 1000 patients with coronavirus disease 2019 in New York: retrospective case series.* BMJ (Clinical research ed.), 2020. **369**: p. m1996-m1996.
- 47. AABB, A., ARC, ASBP, Circular of Information for the Use of Human Blood and Blood Components. 2017.
- 48. Roubinian, N.H. and D.J. Triulzi, *Transfusion-Associated Circulatory Overload and Transfusion-Related Acute Lung Injury: Etiology and Prevention.* Hematology/Oncology Clinics of North America, 2019. **33**(5): p. 767-779.
- 49. Kuldanek, S.A., M. Kelher, and C.C. Silliman, *Risk factors, management and prevention of transfusion-related acute lung injury: a comprehensive update.* Expert Rev Hematol, 2019. **12**(9): p. 773-785.
- 50. Saadah, N.H., et al., Comparing transfusion reaction risks for various plasma products an analysis of 7 years of ISTARE haemovigilance data. British Journal of Haematology, 2018. **180**(5): p. 727-734.
- 51. Eroshenko, N., et al., *Implications of antibody-dependent enhancement of infection for SARS-CoV-2 countermeasures*. Nature Biotechnology, 2020. **38**(7): p. 789-791.